Author's response to reviews

Title: Chronic Depression: Development and Evaluation of the Luebeck Questionnaire for Recording Preoperational Thinking (LQPT)

Authors:

Tanja Kühnen (kuehnen.tanja@web.de)
Franziska Knappke (fknappke@ahg.de)
Tanja Otto (taniaotto@web.de)
Stephanie Friedrich (Stephanie.Friedrich@psychiatrie.uk-sh.de)
Jan P Klein (Philipp.Klein@psychiatrie.uk-sh.de)
Kai G Kahl (kahl.kai@mh-hannover.de)
Michael Hüppe (hueppe@uni-luebeck.de)
Valerija Sipos (Valerija.Sipos@psychiatrie.uk-sh.de)
Ulrich Schweiger (Ulrich.Schweiger@psychiatrie.uk-sh.de)

Version: 3 Date: 16 December 2011

Author's response to reviews: see over
Dear Editors,

many thanks for accepting our manuscript for publication in your excellent journal.

We have made the following changes:

1. We have made a correction in the style of the list of authors in the manuscript. They are now exactly as they are in the submission system. In our final version of the manuscript we use your preferred style.

2. References: We use in our final version of the manuscript squared brackets and do not superscript numbers. All references are cited in the text and we use consecutive numbers.

3. Abstract: This section has the section heading “Abstract” in our final version of the manuscript.

4. Authors Contributions: We use in our final version of the manuscript authors initials.

5. Tables: We removed all visible vertical lines.

6. We have checked our final version of the manuscript for typographical errors.

No other changes were made. We hope that our final version of the manuscript now conforms to journal style.

Sincerely Tanja Kühnen